Author’s response to reviews

Title: An association between hOGG1 Ser326Cys polymorphism and the risk of bladder cancer in non-smokers: a meta-analysis

Authors:

Changwei Ji (dr.jichangwei@gmail.com)
Zhao Liu (dr.zhliu@hotmail.com)
Huimei Chen (chenhuimeii@nju.edu.cn)
Hongqian Guo (dr.guoqian@gmail.com)
Changjian Liu (dr.cjliu@hotmail.com)

Version: 4 Date: 2 July 2012

Author’s response to reviews:

June 30, 2012

Dear Mr Danrolf de Jesus,

Thank you for your email and kind suggestions. We have revised the manuscript, and would like to re-submit it for your consideration. We have addressed the comments made by the reviewer and attached them to this letter. All the amendments are highlighted in red in the revised manuscript.

Looking forward to your reply.

Yours sincerely,
Changjian Liu, M.D.

321 Zhongshan Road

Department of Vascular Surgery

The Affiliated Drum Tower Hospital

Nanjing University Medical School

Nanjing 210008, China

Phone: 86 13951933019

Fax: 86-25-83105107

Email: dr.cjliu@hotmail.com
Replies to Reviewer

The authors have not adequately addressed the concerns I raised. They should have at least acknowledged the key limitations of their findings that had been raised in my first review. And use of random-effects models and Funnel plots methods does not solve the potential publication bias issue.

Response: We would like to express our sincere thanks to the reviewer for the constructive comments. Additional discussions have been added in the revised version (Page 10, paragraph 1) and the limitations of our study have been addressed as suggested (Page 10, paragraph 3).